Author's response to reviews

Title: Biological efficacy of low versus medium dose aspirin after coronary surgery: results from a randomized trial [LREC 01/624]

Authors:

Eric Lim (eric.lim@cvsnet.org)
Jacqueline Cornelissen (Jacqueline.Cornelissen@papworth.nhs.uk)
Tom Routledge (Tom.Routledge@papworth.nhs.uk)
Ayyaz Ali (Ayyaz.Ali@papworth.nhs.uk)
Stephen Kirtland (Stephen.Kirtland@papworth.nhs.uk)
Linda Sharples (Linda.Sharples@papworth.nhs.uk)
Kate Sheridan (Kate.Sheridan@papworth.nhs.uk)
Sarah Bellm (Sarah.Bellm@papworth.nhs.uk)
Helen Munday (Helen.Munday@papworth.nhs.uk)
Stephen Large (Stephen.Large@papworth.nhs.uk)

Version: 2 Date: 5 December 2005

Author's response to reviews:

Manuscript now contains Protocol ID number from the US NIH trials registry (www.clinicaltrials.gov)